

Format for ANSWERING REVIEWERS



August 25, 2012

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 3546-review.doc).

Title: Nanoparticle Based Inner Ear Therapy

Author: Ilmari Pyykkö, Jing Zou Ya Zhang, Weikai Zhang, Hao Feng, Paavo Kinnunen

Name of Journal: *World Journal of Otolaryngology*

ESPS Manuscript NO: 3546

The manuscript has been improved according to the suggestions of reviewers:

- 1 Format has been updated
- 2 References and typesetting were corrected
- 3 Copyright assignment from each partner

Sincerely yours,

A handwritten signature in blue ink, appearing to read 'Ilmari Pyykkö', with a stylized flourish at the end.

Ilmari Pyykkö
Professor and Head
Department of Otolaryngology
Tampere University Hospital
P.O. Box 2000
Teiskontie 35
FIN-33521 Tampere
Tel: +358-3-311 66387
Mobil 358-40-529 4314
Fax 358-3-311 64366